MAT203C University of California, Davis Spring 1998

GENERAL INFORMATION

Instructor:

Bruno Nachtergaele Office: 669 Kerr Hall Office Hours: Mon 11:00–12:00 am E-mail: bxn@math Wed 11:00–12:00 am Phone: 752-8061 Fri 11:00–12:00 am or by appointment

Time and Place:

MWF 3:10 – 4:00 pm, 212 Wellman

Text:

We will closely follow the lectures notes by John Hunter and myself, which will be available on the web at http://www.math.ucdavis.edu/~bxn/m203 public/m203.html. I will assume you have a copy of these notes, e.g., when assigning homework problems. Often, at the beginning of a chapter, references for further reading are given. The most important ones of these are on a three-day reserve at the Shields library. Analysis, by Lieb & Loss is recommended additional reading for this course.

Homework assignments:

There will be weekly homework for this course. The homework is due the next Monday at class time, unless otherwise arranged. The assignment for the week will be posted on the homepage.

TA, Work Sessions:

Sergio Lucero is the TA for this course. He will meet with you once a week to work on problems related to the material of the week. These discussion sessions are also an ideal opportunity to ask questions about the material, or to go over arguments that need clarification. Participating in this session is worth 1 credit unit. But you have to enroll separately for it to get this unit. The course number for the discussion is MAT290-025 and the CRN is 67209. The CRN of MAT203C itself is 67171. Time and place of the discussion session: TBA during the first class meeting on Friday April 3.

Tests and Exams:

This quarter exams will be take-home and open-book. 1. A midterm exam will be given during Week 6 (05/11-15).

2. A comprehensive final exam will take place in the final exam period (06/13-19).

Grade components: - Homework: 20% - Midterm: 30% - Final: 50% Enjoy the quarter, Bruno